Geoforum 117 (2020) 246-258



Contents lists available at ScienceDirect

# Geoforum

journal homepage: www.elsevier.com/locate/geoforum





# Social forestry in Southeast Asia: Evolving interests, discourses and the many notions of equity

Grace Y. Wong <sup>a,\*</sup>, Moira Moeliono <sup>b</sup>, Indah W. Bong <sup>b</sup>, Thuy Thu Pham <sup>b</sup>, Muhammad A. K. Sahide <sup>c</sup>, Daisuke Naito <sup>d</sup>, Maria Brockhaus <sup>e,f</sup>

- <sup>a</sup> Stockholm Resilience Centre, Stockholm University, Kräftriket 2B, 10691 Sweden
- <sup>b</sup> Center for International Forestry Research, Jalan CIFOR, Situ Gede, Sindang Barang, Bogor 16115, Indonesia
- <sup>c</sup> Faculty of Forestry, Universitas Hasanuddin, Jl. Perintis Kemerdekaan KM.10, Tamalanrea Indah, Makassar, Sulawesi 90245, Indonesia
- <sup>d</sup> Graduate School of Agriculture, Kyoto University, Kitashirakawa Oiwake-cho, Sakyo-ku, Kyoto 606-8502, Japan
- e International Forest Policy, Department of Forest Sciences, University of Helsinki, Latokartanonkaari 7, 00014 Helsinki, Finland
- f Helsinki Institute of Sustainability Science (HELSUS), University of Helsinki, Yliopistonkatu 3, 00014 Helsinki, Finland

#### ARTICLE INFO

#### Keywords: Social forestry Assemblage Equity Actors and interests Southeast Asia

#### ABSTRACT

Southeast Asia has long promoted social forestry (SF) in conservation areas, fallow forests, tree plantations, areas in timber concessions and locally managed agro-forest systems, with the engagement of diverse actors and objectives. SF has evolved from early aims of empowerment and devolution of rights advocated by global reform movements, and is now reframed in the market ideal as a win-win-win endeavor for sustainable forest management, climate change mitigation and robust entrepreneurial livelihoods. Southeast Asian states have formulated numerous standardized SF programs and policies that are often linked to broader development goals and priorities, but which have not always been a 'win' for local communities in falling short to provide full tenure rights. Civil society organizations that have provided grounded perspectives on environmental justice and rights have also converged with states on entrepreneurship and market-based solutions. Meanwhile, the private sector actor that is seen as key to these solutions is conspicuously absent within the SF policy space. Within this space of diverse and at times contradictory objectives, whose interests do SF policies serve? We examine the social forestry assemblage to investigate the different discourses, interests and agendas in the implementation of SF schemes in Indonesia, Vietnam, and the Malaysian state of Sabah. The formal SF schemes involve shifting or reinforcing old discourses around forest problems and possible solutions, territorialization processes that can lead to inequities in the exclusion of rights, participation and access, and risks exacerbating contestations and inequities in claims to forest land and resources.

#### 1. Introduction

Southeast Asia is currently pushing forward a revived social forestry agenda. Indonesia has set a target of 12.7 million hectares (mil ha) of forest, approximately 10% of the state forest estate, to be managed under various formal social forestry schemes by 2019, and official statistics suggest that the Indonesian Ministry of Environment and Forestry (MOEF) has allocated permits for 3.4 mil ha as of September 2019. Vietnam set a target of 4 mil ha for social forestry which it met in 2016, four years ahead of its target date (RECOFTC 2017) and of which, most

are being managed by commune people's committees (MARD 2017). In Malaysia, Sabah's Forest Policy 2018 emphasizes sustainable forest management (SFM) principles to generate diverse revenue streams for the state and benefits local communities (Sabah Forestry Department 2018), with social wellbeing objectives that aim "to strengthen the participation local communities in forest management activities" and "to encourage Native [people's] participation in forest industries" (ibid, p. 31).

Social forestry (SF) is broadly defined as "initiatives, sciences, policies, institutions, and processes that are intended to increase the role of

<sup>\*</sup> Corresponding author.

E-mail address: grace.wong@su.se (G.Y. Wong).

<sup>&</sup>lt;sup>1</sup> We use the term social forestry as this is commonly used in policy documents through much of Southeast Asia. in this context social forestry is often used interchangeably with community forestry, community based forest management and similar terms, and we adopt this wide conceptualization for the purpose of this paper.

local people in governing and managing forest resources" (RECOFTC 2014), and includes formalized customary and indigenous governance as well as government-led initiatives. However, the basic tenets of social forestry schemes are the devolution of rights to local communities, support to livelihoods and alleviation of poverty and conservation or sustainable management of forests (Gilmour 2016, RECOFTC 2014). While earlier framings of social forestry policies and initiatives in the region emphasized devolution of tenure rights and local empowerment in forest management and conservation, these have evolved into SF as a model for agrarian transformations, and as a foundation to support neoliberal solutions as reflected in the trend towards community forest entrepreneurship, commercialization of forest products and private sector investments (MacQueen 2013, Moeliono et al. 2015). More recently, social forestry initiatives are being reframed as a solution to support climate change mitigation and ecosystem service provision (Agrawal and Angelsen 2009, Hajjar et al. 2016).

Formal social forestry schemes with their specific objectives and implementation models are being implemented in rural forested regions where demography and livelihoods are rapidly transforming with increased mobility and migration (Kelley et al. 2019; Peluso and Purwanto 2018), increased cultivation of cash crops and rising non-farm incomes (Rigg 2003), changing perceptions and expectations on communal versus individual tenure (Fisher et al. 2018, Moeliono et al. 2016) and changing perceptions of the roles and values of ecosystem services from forests (Pham et al. 2012). Against this backdrop of change, we draw on the notion of assemblage that builds on framings in forestry (Li 2007, Nel 2017, Astuti and McGregor 2017) to tease out the different actors and their interests, and the ways in how the material and discursive shape their interactions in case studies from Indonesia, Vietnam and the Malaysian state of Sabah. We argue that current assemblages of social forestry have created more losers than winners. The dominant interests and materialities that emerge from messy interactions in the social forestry space have been translated as institutionalization of restrictive rules around access and rights, and discourses of market-based approaches that limit local agency and imaginations of sustainable futures. As formal social forestry schemes unfold within the specific social, ecological and political contexts, we examine how discourses in forestry - the framing of problems and their solutions - shift, accompanied by changes in the design of social forestry policies and their notions of equity and empowerment.

The paper is structured as follows: Section 2 provides a historical context to the development of social forestry in Southeast Asia and the theoretical frame on assemblage. After a brief discussion on data and methods in Section 3, Section 4 presents the three case studies highlighting how discursive problematizations of forest management over time in each of the regions leads to changing policies and how social forestry is posited within the set of 'solutions'. Section 5 discusses the SF assemblage and how specific agendas and interests are served by the formal social forestry models in the case studies, and how processes of territorialization affects notions of equity. Section 6 concludes with some perspectives for future SF research in Southeast Asia.

# 2. Background and theoretical frame

### 2.1. State and forestry in Southeast Asia

Forests is a contested resource and political space in Southeast Asia with many having staked claims on the resource across its history. Forest dwellers and smallholders have long lived on goods and services from forests. The colonial urge to control and produce valuable commodities from forests are still dominant in Southeast Asian states' land and forest use policies (Peluso, Vandergeest and Potter 1995). The Dutch colonial government established teak plantations that cover about 8% of Java island's land area and nearly a quarter of the island is state controlled forest of one kind or another (Peluso 1992). The Dutch also initiated the establishment of the 'Forest Estate' (Bosch Wezen) with enforced

boundaries and exclusion policies. State control over forests has now expanded to almost two thirds of Indonesia land, or 126 mil ha, under the control of the Ministry of Forestry in the form of State Forest Estate (Fisher et al. 2018, Siscawati et al. 2017). In its centralized role, the Indonesian state has identified and classified forests and land based on technical, bureaucratic and institutional practices of forestry (Peluso and Vandergeest 2001).

After independence from French colonial powers in 1945 which had established instutions for exploitation of forest resources and development of a plantation economy (Cleary 2005), Vietnam established a socialist form of state forestry with centralized state management and creation of state forestry enterprises for the continued 'rational' exploitation of forest (Dang, Turnhout and Arts 2012). In the Malaysian state of Sabah (formerly British North Borneo), the North Borneo Chartered Company claimed ownership of all land, forests and resources in order to generate revenue for British shareholders during its reign from 1877 to 1946<sup>2</sup> (Doolittle 2007). The colonial legislation was adopted as part of Sabah's independence package and in the formation of Malaysia together with Sarawak and the Federation of Malaya (Lunkapis 2013).

Thus in Indonesia, Vietnam and Sabah, post-independence forest policies were implemented with the same rationale of economic production and little consideration of the diverse social-cultural and economic needs and practices of local, forest-dependent communities. Local forest practices and land uses were depicted in outdated colonial ideologies as "@" backward, lazy and responsible for the past and current state of deforestation and resource degradation" (Doolittle 2007, p. 476). Shifting cultivation practices in particular were considered as a threat to both, forest conservation and production, (Thu et al., in press) to be banned and replaced with more productive – and profitable – forms of agriculture (Cleary 2005, Peluso, Vandergeest and Potter 1995). The path dependency of such policies and ideologies has had long term consequences for how social forestry has been conceived, with tangled legacies and persisting contestations over customary rights. In Indonesia an estimated 37-48 million people in over 25,000 villages live within or on the border of Forest Estate lands (MOEF 2018) and have variably claimed customary or individual rights over forests, used forest resources for livelihood and social needs, and converted forests to agriculture. In remote areas where the Vietnamese state has less control, ethnic minority groups circumvent the rules and continue to manage forest resources and practice a range of swidden agriculture systems (McElwee 2011, Thu et al., in press). The colonial government in Sabah has tended to treat local communities and their property regimes as homogenous and rigid institutions (Doolittle 2001), despite studies demonstrating a long history of diverse and nuanced forms of rights to forest products, trees or water (Peluso 1996), a notion repeated in recent implementation of policies on native land titling. Complexities around the dynamism and pluralism of social and political institutions, interests and power relations governing forest access and control (Leach 2002) are typically buried and simplified under layers of bureaucratic governance and expectations for the potential of social forestry.

## 2.2. The social forestry assemblage

The global push towards community-based, and participatory approaches to conservation and sustainable forest management in the early 1990s advocated for devolution of rights to local communities for the responsibility to manage natural resources held as commons (De Jong, 2012, Moeliono et al., 2017). More recently, new global–local and public–private alliances are shaping the negotiation of agendas and interests, with forests being seen as stores of global biodiversity and

<sup>&</sup>lt;sup>2</sup> The British reign was interrupted from 1941 to 46 with the Japanese occupation during World War 2. North Borneo then became a British crown colony from 1946 until its independence in 1963.

carbon wealth, sources of ecosystem services or timber value, and as resources to support local livelihoods or cultural concerns. In exploring the interplay between these agendas, the old discourses, in which outdated knowledge is entwined with power and institutional interests, has emerged to bring back into play older attitudes around around rights, shifting cultivation and community institutions (Skutsch and Turnhout 2018).

We put forward an assemblage approach as it allows for an understanding how the diverse interests, agendas and discourses around social forestry can cohere in a particular place. We do so by building on Li's (2007) definition of the social forestry assemblage as "things (trees, logs, non-timber forest products, tools, documents), socially situated subjects (villagers, labourers, entrepreneurs, officials, activists, aid donors, scientists), objectives (profit, pay, livelihoods, control, property, efficiency, sustainability, conservation) and an array of knowledges, discourses, institutions, laws and regulatory regimes" (Li 2007, p. 266). More recently, these objectives have included climate mitigation and adaptation and rural development. The assemblage approach highlights possibilities of emergence, cohesion, shifts, dispersion, and multiplicity (Anderson and McFarlane 2011, Cons and Eilenberg 2019). Seen as sprawling collections of messy interactions between socio-material elements, assemblage is a useful approach for connecting the complexities of the political nature of state-local relations in social forests.

Assemblages are also characterised by processes of territorialisation; and forest boundaries are delineated through colonial and contemporary policies of gazettement for production or conservation, formation of social identity or claims, or arrangement of jurisdictional and administrative borders (Cleary 2005, Peluso and Vandergeest 2001). Territorialisation can signify exclusion such as in evictions from contested protected areas under new carbon forestry projects (Nel 2017) or omissions of certain ethnicities in a process of producing homogeneity to which particular community comes to declare itself as a specific Indigenous entity (Astuti and McGregor 2017). Similarly, social forestry with claims to "fix" forestry governance problems or counter exclusions of locals from forests and land, come with formal boundaries and territorialization processes that will inevitably exclude those that cannot perform the type of stewardship required or do not fit within predefined notions of "community".

The social forestry assemblage varies with context but share commonalities in how disparate objectives and problem framing is pulled together in relation to "diagnoses of deficiencies and promises of improvement" (Li 2007, p. 264). The discursive problematization of forestry and forest management is not static but changes over time (Moeliono et al. 2017, De Jong 2012, Peluso, Vandergeest and Potter 1995) and along with these changes, are shifting promises of improvement or proposed solutions. The central proposition of social forestry is that local people living in and close to forests can best manage forests effectively over the long-term. This 'solution' does not however come from a unified specification of a problem. Local people were once considered to be the culprits in driving deforestation and degradation but are now seen to be the solution to effective forest conservation (Gilmour 2016, Peluso, Vandergeest and Potter 1995). They have shifted from being the subjects of sustainable forest management to agents for implementing various sustainability schemes under the umbrella of social forestry (Skutsch and Turnhout 2018).

We lean on assemblage theory to tease out the different actors and their negotiations and adaptations of interests in the design and implementation of social forestry mechanisms. As their seemingly disparate and heterogeneous objectives collide, disperse and realign, we assess how the deficiencies in social forestry are problematized at different points in time, how the solutions proposed for their improvement shift or hold constant, and how the processes of territorialization can have direct equity implications. Our specific research questions are thus twofold:

- How do the key actors in the SF assemblage impose, negotiate, contest and/or align their interests in design of SF policies and territorialization processes at particular times?
- What are the changing notions of equity related to changing discourses around SF, resultant SF solutions (policies, mechanisms and territorialisation processes)?

#### 3. Data and Methods

Data for this paper was collected under a project on social forestry and climate change<sup>3</sup>. As part of the project's activities between 2012 and 2019, we've carried out reviews of national policies and published and grey literature related to social forestry, conducted interviews with policy and civil society actors across different levels of governance, and carried out field research in Vietnam, Laos, Indonesia and Malaysia Sabah<sup>4</sup>. From this work, a series of articles related to diverse local forest governance systems and on the intersections of social, economic and political dynamics within forest-agriculture frontier systems in Southeast Asia have been published (see for example Bong et al. 2019, Cole et al. 2019, 2017, Maharani et al. 2019, Moeliono et al. 2017, 2016, Wong et al. 2019, and Thu et al., in press). In addition, the authors have engaged in policy discussions related to social forestry at the national level with the state-led Vietnam Community Forestry Working Group, networks such as Indonesia's Working Group on Social Forestry and the Sabah Social Forestry working group (SASOF), and participation in regional policy events such as the annual ASEAN Working Group on Social Forestry and ASEAN Senior Officials in Forestry meetings from 2012 to 2019.

For this paper, we leveraged on findings from reviews of national policies, interviews conducted with policy and civil society organizations, and our observations in national and regional policy events. We also carried out new analyses of strategic plans and policy materials published by key actors within the social forestry policy arena, and ASEAN policy documents (see Table 1). The key actors were identified through their regular and active engagement in ASEAN and national social forestry events. Within the data corpus, we have identified the actors' interests and discourses through a close reading of supported policies and vision statements and objectives in strategic plans; their rhetoric in how the problem of forests and people are framed in the documents' background and contextual sections; and how this is linked to the proposed actions or solutions for improvement (see Table 2).

We describe the three case studies in the following section, highlighting in particular the problem and solutions identification, actors, interests and agenda setting within the social forestry assemblages in Indonesia, Vietnam and Sabah.

# 4. Case studies of social forestry schemes

In all three case studies, social forestry is being implemented within boundaries of conservation areas, forest reserves, village forests, fallows and agroforest systems, areas around tree plantations and timber concessions, which suggests different materialities, politics and exclusions.

Social forestry took hold in Indonesia in the early 1990s in line with the global ideals of community based management being promoted through projects funded by international organizations such as the Ford Foundation (Seymour and Rutherford, 1993). The Government of

<sup>&</sup>lt;sup>3</sup> The project ASEAN-Swiss Partnership on Social Forestry and Climate Change, was funded by Swiss Agency for Development and Cooperation and implemented by the Center for International Forestry Research (CIFOR) and partners to support ASEAN policy-making. This paper derives from the research carried out solely by CIFOR, see cifor.org/ASFCC. Funding for the research in Sabah was provided by the Ministry of Foreign Affairs, Japan.

<sup>&</sup>lt;sup>4</sup> Whilst field research and policy interviews in Vietnam and Indonesia started in 2013, field research in Malaysia Sabah only started in 2018.

Table 1
Review of key actors' policy and strategic planning documents.

Key actors	Scope of influence	Category	Policy documents
ASEAN Working Group on Social Forestry	Southeast Asia wide	Policy actors	1. Vision and Strategic Plan for ASEAN Cooperation in Food, Agriculture and Forestry (2016–2025) 2. Strategic Plan of Action for ASEAN Cooperation on Forestry (2016–2025)
Center for People and Forests (RECOFTC)	Southeast Asia wide	Civil society organization	3. Resilient forest landscapes: Empowered communities, strengthened institutions and shared prosperity (Strategic Plan 2018–2023) (RECOFTC, 2018)
Non-Timber Forest Products- Exchange Program (NTFP- EP=	Southeast Asia wide	Civil society organization	4. Linking People and Forests: Strategic Direction 2016–2019 (NTFP-EP, 2016)
Ministry of Environment and Forestry (MOEF)	Indonesia	Government	5. State of Indonesia's Forest 2018 6. Ministerial decree P.84/MENLHK. SETJEN/2015 on Rights Forest 7. Ministerial decree P.83/MENLHK/ SETJEN/KUM.1/10/ 2016 on Social Forestry
Forum for Communication on Community Forestry (FKKM)	Indonesia	Civil society organization	8. Warta Edisi Augustus 2010 (FKKM (Forum Komunikasi Kehutanan Masyarakat), 2010) 9. Warta Edisi November 2017 (FKKM, 2017)
Working Group on Social Forestry	Indonesia	Multi- stakeholder group, formalized by MOEF	10. Decree of the Director General of Social Forestry: P.14/PSKL/SET/ PSL.015/2016 on the establishment of the Social Forestry Acceleration Working Groups (POKJA PPS)
Sabah Forestry Department (SFD)	Sabah	Government	11. Sabah Social Forestry Roadmap (as part of the Malaysia Social Forestry Roadmap), unpublished draft (Oct 2017) 12. Sabah Forest Policy 2018 (Sabah Forest Department, 2018)
Ministry of Agriculture and Rural Development (MARD)	Vietnam	Government	13. Vietnam Forestry Development Strategy (2006–2020) 14. Political Report of the Communist Party of Vietnam's 10th Central Committee to the 11th National Party Congress 2011

Table 1 (continued)

Key actors	Scope of influence	Category	Policy documents
			15. Forestry Law 2017 16. Decision No. 1565/ QD-BNN-TCLN in 2013 approving the forestry sector re- form proposal
			17. Decree No. 02/ 1994/ND-CP on the enactment of regulations on the allocation of forestry land to organizations, households and individuals for stable and long term use for forestry purposes
			18. Decree No. 01/ 1995/ND-CP on land allocation for use in agricultural, forestry production, aquatic product rearing within state-owned enterprises

Indonesia did not initially consider social forestry a policy for decentralization of rights but rather, as a "development" program for villages located in forested areas where forests had been used or managed in some form or other (Moeliono et al. 2017). SF was formally included in the new Forestry Law of 1999 and in subsequent policies and laws. Indonesia's conceptualization of social forestry today is legalized in five distinct schemes or permits, each with its specific management regime, permitting requirements, eligible applicants and extent of rights allowable in specific forest zones (Erbaugh 2019, Moeliono et al. 2017).

Vietnam moved towards social forestry in the 1990s as forest exploitation led to a rapid decline of forest resources. The implementation of a logging ban to address this problem resulted in a financial crises for most state forest enterprises and a collapse of the centralized forest management. The 1993 Land Law and trade liberalization under the *Doi Moi* socio-economic reforms promoted 'forest socialization', a Vietnamese term that emphasizes the contributions of societal stakeholders to achieve good forest governance (Dang, Turnhout and Arts 2012). The 2004 Law on Forest Protection and Development legalized the allocation of natural forests to community management and encouraged local people to increase their revenues through community forestry (Pham et al. 2012) despite restrictions on use. The Forestry Law 2017 further acknowledged religious and customary forms of forestry but the Land Law 2013 continues to provide only limited rights to local communities

Sabah has carried out intensive, industrial-scale logging since the 1950s, reaching its peak in the early 1980s. During this period, almost 90% of the state's budget was derived from timber revenue. Social forestry was first introduced by the Sabah Forestry Department (SFD) in the early 1980's to address issues of "illegal" settlement and swidden cultivation in forest reserves (Toh and Grace, 2005) despite the fact that communities would have lived in the area before their designation as forest reserves and had customary rights or claims to the land (Lunkapis 2013). SF projects were initially funded through Community Forestry Cess Fund, established in 1989, through fees collected from timber companies based on the volume of timber exported or processed (Toh and Grace, 2005) and have become a consideration for sustainable forest management licence agreements (SFD Annual Report 2016).

# 4.1. Case study 1: Social forestry and agrarian reform in Indonesia

The presidency of Joko Widodo, which began in 2014, introduced

**Table 2**Key actors and their interests, rhetoric of the problem at hand and proposed actions, based on review of social forestry related policy documents. <sup>a</sup>

Actor	Core interests	Problematization	Proposed actions and solutions
Regional actors: ASEAN	"A competitive, inclusive, resilient and sustainable Food, Agriculture and Forestry (FAF) sector integrated with the global economy, based on a single market and production basse contributing to food and nutrition security and prosperity in the ASEAN Community." [1, p.6]	" in particular, deforestation, largely driven by increasing demand for agricultural land and aggravated by illegal logging. These developments now threaten ASEAN's enviable current position as a region with high levels of food security and a net exporter of food, industrial crops and forestry products." [1, p.5]	Social forestry should be designed to "effectively contribute to sustainable forest management at the local level, climate change adaptation and mitigation, rural development and poverty alleviation, as well as improved livelihoods and social well-being" [2, p.15]
	Priority for Forestry sector is sustainable forest managment [1, p. 14] with emphases on market mechanisms (certification, value chains) and leveraging on REDD+; PES and climate finance.  Sustainable forest management as a triple win for production of forest goods and services,	Social forestry to provide alternatives to "deforestation driven by basic needs, such as subsistence farming and the reliance on fuelwood for energy" [2, p.15]	SF is expected to "review customary and statutory tenure arrangements at the national level, including access and use rights of indigenous people, local communities, forest communities and other forest-dependent communities as provided for in the UN Declaration on the Rights of Indigenous Peoples (UNDRIP) of 2007 [2, p.15]
RECOFTC	biodiversity conservation and their optimal use, and climate change mitigation/ adaptation [2] "Envisions a world where empowered local people are effectively and equitably engaged in the sustainable management of forest landscapes" [3, p.8]	Economic development and integration impacts the poor disproportionately: "local people are often cut off from the benefits of development but bear the greatest risks. Many people – especially women, indigenous peoples and other vulnerable populations – have had their lives upended by	Through applying Community Forestry: local livelihoods are increasing, forests are more sustainably managed, deforestation reduced, conflicts minimized, greater equity is achieved and landscapes are better able to withstand natural disasters and climate change [3, p.14]
		forces far beyond their control and millions more are at risk of displacement, strife, poverty, conflict and tragedy" [3, p.8] "Development brings greater risk of land-grabbing and increases the pressure on land, often with little regard for safeguards" "As development progresses, conflicts around forest landscape management are becoming more common, more complex and more severe" "Along with rapid development and regional integration, the stakes and opportunities for private businesses are rising as investments flow in and out of countries. Investments may have serious impacts	Focus is on building/enhancing landscape collaboration in a changing climate; strengthening good governance, institutions and conflict transformation; private sector engagement and enterprising communities; enabling social inclusion, gender equity and public action.
NTFP-EP	"— an effective way to protect and enrich the forests, and simultaneously fight poverty, is by helping forest dwellers make a good living from the forest itself by increasing the value of the various non-timber forest products (NTFPs)" [4, p.10]  "Envisions self-empowered generations of forest-dependent communities contributing to	on forest landscapes and communities" [3, p.10] "Conversion to agriculture, including the recent expansion in the region devoted to oil palm plantations, continues to be the main cause of forest loss particularly in Southeast Asia There are underlying causes of forest loss and degradation including undervaluing the ecological and socio- cultural functions of forests, national development priorities on agro-industrial development, poor natural resource tenure framework and elite	Strategic actions include: strengthening community ecological monitoring, engagement with governments on community forest tenure rights, facilitate community based enterprises and develop value-base and sustainable market linkages, assess and advocate for wild foods, nutrition and food policies. Focus on cultural values and gender equality [4, p.20–27]
Country actors:	and benefiting from sustainably-managed forested landscapes and ecosystems, knowled and cultural traditions" [4, p.15]	"The vulnerable poor who live in or near forests are often the most affected by deforestation. This situation is compounded by global warming, social inequities, lack of forest tenure rights, and an ever expanding consumer market which radically heighten the vulnerabilities of these forest-dependent communities." [4, p.8]	"to help shape and influence policies that promote more efficient and cost-effective NTFP production. It will tap networking options for communities and for their products and enterprises.  Biodiversity-friendly and ethical green products will be promoted for the niche markets and tourism will tap more strategic market opportunities, but at the same time advocate for ethical domestic, regional and international markets for NTFPs and creative." [4, p.17–18]
Indonesia Ministry of Environment and Forestry Indonesia	In order to " reduce poverty, unemployment and inequality in the management / utilization of the forest, a Social Forestry activity is needed to provide legal access to the local community in the form of Community Forests, Village Forest, Forestry Partnerships and Community Plantation Forest" [7, Art.1].	Lack of access to land considered as a main cause of poverty [5, 7].  "Since the 1970 s, problems related to poverty, land conflict and forest degradation have affected forestry management" [5, p.83]	Social forestry "provides solutions to unemployment, to poverty, to land conflicts, for the rehabilitation of lands and restoration of landscapes, and provide a sense of security and peace of mind to communities by providing them with legal of access to forest resources and the Forest Area" [5, p.86]. The four formal SF schemes provide limited use and/or management
	In addition, the Ministerial decree on SF "seeks to resolve the tenure of justice and justice for the local community and the customary law community within or around the forest area in the framework of community welfare and forest function conservation" [7. Art. 2.2].		rights for a specific timeframe through a permitting process [7]. Customary or Adat Forest provide tenurial rights through a multi-level legislative process [6].

(continued on next page)

function conservation" [7, Art. 2.2].

Table 2 (continued)

Actor	Core interests	Problematization	Proposed actions and solutions
FKKM	A reorientation of SF from a focus on community empowerment to one of community welfare [5]. Established in 1997, FKKM's vision is to forward the perspective that the community forestry organization is "based on the principal of justice, transparency and responsibility, as well s ecological, economic, and sociocultural sustainability"  It's mission is to motivate the development of CF in this perspective through information dissemination, concept development, capacity building and policy formulation. [8, 9]	Community forestry is one solution to the problems of tenure and structural poverty to the destruction of forest resources.  Social Forestry has yet to address the fundamental problems faced by the customary and local communities. The perpective that must be changed into a new 'Community Forestry' perspective followed by decentralization and evolution, access to forest resources for customary and local communities, adoption of the forest resource and	FKKM strategy is to extend community forestry and shared learning process; support conflict mediation; develop the means for information exchange and prmotion of community forestry; and facilitate policy reforms to meet the principals of community forestry [9]
Working Group on Social Forestry (WG-SF)	The WG-SF is a multi-stakeholder platform, formalized by the 2016 government decree on Social Forestry Acceleration Working Groups [10].  As part of the official SF strategy, every province need to establish a Social Forestry Acceleration Working Group (POKJA PPS) consisting of government, NGOs, academia, and interested parties [10].	forest ecosystem management systems. [8] The WG-SF supports the government's narrative as it is tasked to raise awareness, and support local governments and local communities in the verification and formal processes required to obtain SF permits.	Mobilizes POKJA PPS at the provincial level, with responsibility to the provincial governors [10].  Enables a transparent verification and monitoring process.
Country actors: Sabah Sabah Forestry Department (SFD)  Country actors:	Mission of SFD is to "effectively and efficiently plan and implement the management of the State's forest resources in accordance with the principles of sustainable forest management (SFM)" [12, p.6]  Commitment to ensure at least 50% of land is "designated and protected for sustainable forest use and tree cover for environmental protection, biodiversity conservation and socio-economic well-being." "The Forest Policy strengthens the commitment and direction in managing designated forest areas and tree cover through sustainable forest management. This policy takes into account environmental, social and economic sustainability, through good forest governance and best management practices, to ensure forestry remains an integral and competitive land use in Sabah." [12, p.7]	"With the decline in wood production from natural forests, planted forests have become an important supply of timber domestically for the futuretap on new sources of forest revenue from carbon, PES and NTFP" [12, p.20]	Aims of economic wellbeing policy thrust is to ensure sustainability and viability of the wood industry; optimize forest rents and revenues from PES, NTFPS, nature-based tourism and climate finance. [12, p.28–30] Aim of social wellbeing policy thrust is to strengthen participation of local communities in forest management activities through increased consultations and dialogues; preferences for local and native employment and contracting; native entrepreneurship [12, p.31]
Vietnam Ministry of Agriculture and Rural Development	Social forestry is seen as a mechanism to engage local communities to manage forests, support poverty reduction and generate financing to the forestry sector.  " there is a need to make comprehensive adjustments for the orientation of sector development in order to meet the needs of renovation and the trends of international economic integration, thereby creating conditions for the mobilization of more domestic and foreign resources for sectoral development investments" [13, p. 2]. Forestry Law 2017 recognizes ethnic minorities in forest land ownership for the first time: "The State shall allocate forest and land to ethnic minority people and communities whose income mainly comes from forests for combined forestry-agricultural-fishery production; facilitate cooperation in forest protection and development with forest owners and benefit sharing arising from forests; facilitate practice in culture and beliefs associated with forests according to the Government' regulations" [15, Article 4].	The lack of finance to support State forest management has hampered development of other related economic sectors.  "Economic development is not sustainable; quality, efficiency and competitiveness remain low and incommensurate with the potential, opportunities and development requirements of the country; certain targets (including forests) have not been reached" [14]  "New policies and mechanisms have not been supplemented on time for investment in development of production forests, wood processing and NTFPs to create the momentum for accelerating various economic sectors, particularly households, community and private enterprises participating in the development of forestry activities" [13, p. 5].  " forestry development has mainly relied on the state budget, and not yet mobilized to the maximum the resources of non-state sectors and environmental services. The investment in the forestry sector and forestry activities is still very low compared with the needs" [13, p. 6].	Continue allocation of land and forests and ensuring food supply to local inhabitants with a view to forest plantation and protection [14]. Forestry actions with local communities is largely an economic approach:  - Create diversified, flexible collaboration modalities so that farmers and households can contribute their land use right together with enterprises to accumulate land for large-scale forestry production;  - Promote large-scale specialized production zone in the forms of farms and households; develop criteria for household farms, increase by 150% number of forestry households in 2015 and 200% in 2020 compared with that in 2011 [16].

<sup>&</sup>lt;sup>a</sup> The numbers in [] relate to the specific policy documents listed in Table 1.

programs to respond to the idea of developing the nation from its margins. These included agrarian reform programs, of which SF is a major component. This reinvention of SF promoted the idea that by providing access to land, business opportunities and vocational training, poor rural farmers and forest communities will be able to escape poverty (MOEF 2018). Widodo's target of 12.7 million ha of forest area under SF by 2019 has led to an explosion of SF permits. While there are varying figures, official statistics suggest that 1.27 mil ha were allocated to various SF schemes between 2015 to June 2018 (MOEF 2018), with a large jump to 3.4 mil ha by Sept 2019<sup>5</sup>.

This newest iteration of the SF policy is distinct. In the 1970s and 1980s, SF was formulated as a 'development' program for communities living in or around forest plantation concessions to resolve land conflicts and social discontent to further the interests of state and private corporations. This changed in the 1990s and 2000s with a series of ministerial regulations and national forest laws that laid out formal SF schemes for the devolution of rights and access aiming to empower communities to manage forests (MOEF 2018, Moeliono et al. 2017). The current definition of SF is: "a system of sustainable forest management within the state forest area or within privately owned or traditionally (adat) owned forest, whereby local communities or adat communities are the principal actors carrying out forest management to improve their livelihood, maintain environmental balance and their social-cultural heritage in the form of Village Forests, Community Plantation Forest, Community Forest, Traditional [or Customary] /Adat Forest and Partnership Forests" (Ministerial decree P.83, article 1). Officially, the goal of SF remains to solve tenurial conflict and achieve equity for local and adat communities living in or around forests for their wellbeing and sustainability of forests (Ibid, article 2.2).

SF is implemented according to the five formal schemes as per the Ministerial Decree P.83. Three schemes are based on management permits issued to village enterprises (Village Forests), communities/groups (Community Forest) and individual or cooperative plantations (Plantation Forest), one scheme regulates co-management of forests in protected areas (Partnership Forest), and only one recognizes full ownership of forests by customary or 'adat' communities (Adat Forest). However, the process to obtain the certificate of adat ownership involves the laborious process of gaining official recognition as 'adat' community by the district government and a territorialization process that can exclude certain groups and their claims (Myers et al. 2017). Further, adat forests comes with the full responsibility to protect the forest but with little government support to do so (Fisher et al. 2018).

Invisibility of local forest governance practices. Although official SF policies responsibilize local communities to manage forests, participation at the community level is often weak or limited to village elites (de Royer et al. 2018). Further, the SF territory comes with state-imposed restrictions on sites, use and access (De Royer et al. 2018, Erbaugh 2019), thus making impossible for diverse local governance practices and informal rights long practiced in many forested sites to be recognized by, or drawn into, formal social forestry structures (Bong et al. 2019). This incompatibility has created conflicts and contestations over use and access (Cummins and Yamaji, 2019). Despite the restrictions, local communities are still inclined to obtain SF permits for achieve some security of access to the forest land and resources they are dependent on (Myers et al. 2017).

<u>Bureaucracies</u>, <u>politics and interests</u>. To obtain a SF permit, communities have to develop and submit a technical proposal documenting required criteria such as proof of a community institution or cooperative to manage the forest, the boundaries of the area to be managed, and an approved management plan (<u>Moeliono et al. 2017</u>). This territorialization process excludes local groups who may not fit, or choose not to participate, in these "neat" community institutions or cooperatives, or

those who may demand more from or contest the specific discourses of forest stewardship within the formal SF policy (Myers et al. 2017, Sahide et al. 2020). There is also discontent amongst local (district) agencies who used to play substantive roles in forest governance and now excluded under the new Law 14 on Regional Government (Fisher et al. 2018).

In addition to SF, Indonesia's push for agrarian reform is spearheaded by the policy on Identification of Land for Agrarian Reform policy (Identifikasi Tanah Obyek Reforma Agraria, TORA). TORA involves the certification of 4.5 mil ha of land informally managed by individual farmers, the re-distribution of 4.1 mil ha of state forest lands, and the provision of use rights for 0.4 mil ha of idle or abandoned lands (Resosudarmo et al. 2019, KSP, 2016). The latter two categories are forest areas that are earmarked for crop plantations, unproductive forests slated for conversion and potential areas for paddy rice farming (KSP, 2016). This has implications for potential disputes between the MOEF and the Ministry of Agrarian and Spatial Planning (Sahide and Giessen, 2015). SF permits are being awarded or planned for 12.7 mil ha of the State Forest Area while these same areas are also being 'released' as private property to communities and individuals for farming. While contestations over TORA has not provided much space for the involvement of civil society organizations (CSOs), the SF policy process on the other hand has embraced their participation.

A loose multi-stakeholder platform, established since the 1990s of CSOs, academics, individuals and businesses actively engaged in all aspects of social forestry, was institutionalized as the formal 'SF Acceleration Working Groups' (POKJA PPS) through a decree by the Director General of Social Forestry (P.14/PSKL/SET/PSL.0/11/2016) to support fast-track implementation of SF schemes on the ground. This act effectively subdued the platform's critical dissent of the policy by rendering its members as technical experts and agents to implement the SF policies. The POKJA PPS engages in the entire territorialization process of SF at the local and provincial levels, including facilitating permit applications, supporting the administrative and biophysical verification and monitoring processes, and reports to the provincial governors (MOEF 2018, Resosudarmo et al. 2019) again bypassing local government agencies. In interviews, members of the multi-stakeholder platform suggest that priority is to first secure SF permits, and then work on other equity issues later. In addition, SF Community Business Groups were formally established to assist in the development of social forestry enterprises (MOEF 2018), which are considered as a successful outcome of a SF scheme. The Indonesian assemblage with its diverse actors, interests and high levels of interconnectedness and contestations suggest that different possibilities could be negotiated at specific moments and

# 4.2. Case Study 2: A State-driven Social Forestry in Vietnam

The Vietnamese Constitution defines the State as manager of all forests and land,. Insights from interviews with government actors between 2016 and 2018 suggest that social forestry in Vietnam can be understood in two different ways: social forestry (lâm nghiệp xã hội, xã hội hóa ngành lâm nghiệp) and community forestry (quản lý rừng cộng đồng or lâm nghiệp cộng đồng). The term "community forestry" implies attention to communities as the focus of the policies while "social forestry" focuses on mobilizing finance from private sectors and social actor groups. Findings from interviews conducted with central and provincial government actors as well as from reviews of policy documents suggest that the central government emphasizes "social forestry" and forest policies prioritise collaborative models with local communities and households through an economic and production lens. Moreover local governments tend to apply a "community forestry" approach in their forest protection and development plans emphasizing support to and empowerment of farmers, households, and communities to manage forests. The distinction between the two terms suggest differentiated priorities for local actors in the management of the state's

<sup>&</sup>lt;sup>5</sup> https://sinav-perhutanan-sosial.appspot.com/statistika, accessed 2/11/2019.

forest estate.

The forestry sector, as with other sectors, is managed by the State to align with the political interest of the Communist party, whose views towards social forestry are not static. Following independence from the French in 1945, forests were exclusively managed by the State through its state forest enterprises (Directive No. 15 CT/CTCW 1961). The early 1970s saw emergence of a political shift towards collaborative forest management (Forest Protection Act 1972), which led to the coercion of local people to support the state as temporary laborers for state forest enterprises often with no direct benefits in return (Pham et al. 2012). Decision No. 184-HDBT 1982 and Forestry Development Strategy 1986-2009 recognized the role of non-state actors (cooperatives, communities and individual households) to manage forests but without accompanying rights. In the late 1980s, transformation of the country's economy from state central planning to market orientation, Doi Moi, triggered liberalization in the agriculture sector and large-scale changes in household ownership of products and land, but this was not translated to the forest sector with "ownership" still limited to use rights. The processes of territorialization with selective inclusions and exclusions is to facilitate economic production. Significantly, the open door policy of Doi Moi enabled foreign organizations to introduce new narratives and demands for devolution and participation (Dang, Turnhout and Arts

Increasing openness but not for all. Early manifestations of social forestry started in the 1990s (Forest Protection and Development Law 1991 and Land Law 1993) emphasized local communities' duties rather than the benefits they could gain from managing forests (Clement and Amezaga 2008). Many central and provincial government interviewees claimed that "In the past, no one wants to protect and manage forests but central and provincial government assign forests to them as political tasks. Local people saw this as a political mission rather than for their own benefits". In forest areas belonging to state forest enterprises, local people were contracted to work on forest protection or planting (Decree 01/1995/ ND-CP, Decision 661/1998/QD-TTg), with a share of benefits depending on time and labor invested (Ngo and Mahdi 2017). Community forestry was officially recognized in the Forest Protection and Development Law 2004 where "ownership" rights of non-state actors were specified, and the first state-led community forestry pilots were implemented in 17,000 ha across 64 villages in 2007 (RECOFTC 2014). These pilots drew on lessons from the legacy of past forestry policies and projects and the contestations over changing rights and exclusions (Sikor and Nguyen 2011). The government considered community forestry to be a solution to the conflicts between the state and local people in forest and land governance, between conservation and development practices, and an opportunity to address high rate of deforestation and degradation (Decree No. 02/1994/ND-CP, Decree No. 01/1995/ND-CP).

The expansion of collaborative forestry approaches in Vietnam was in large part shaped by the policy narratives of international actors, throughthe influence of bilateral aid programs on social forestry funded by Switzerland (1992-2002) and the Trust Fund for Forest funded by Finland, Netherlands, Sweden and Switzerland (2004-2015). In particular, MARD interviewees (2018) attributed the government's shift towards social forestry evident in the Forest Protection and Development Law 2004 to the Swiss-funded program on social forestry. Declining state budget allocations to the forest sector pushed the government to consider alternative financing from non-state actors (Pham et al. 2008, Pham et al. 2013), leading to the development and implementation of the national Payment for Forest Environmental Services policy (PFES Decision 380 and Decree 99). International donor-funded programs on PES (Germany and the US) and REDD+ programs (Norway) set the tone on the need to engage meaningfully with communities, indigenous people and private sector. However, State dominance in these processes leaves limited political space for non-state actors, e.g., NGOs and CSOs, in Vietnam to exert an influence on the final policy outputs (Pham et al.2014).

The gap between 'community forestry' and 'social forestry'. The Forestry Law 2017 (enacted on 1/1/2019) gave formal recognition of religious and customary rights over forests for the first time in Vietnam's legal framework. While policies have increasingly emphasized the rights and important roles of communities, individual households and private sector as forest owners and managers, this shift remains rooted in the government's interest to ensure sustainable revenues for the forestry sector – thus reflecting the disparity between the Vietnamese meanings of 'social forestry' and 'community forestry'. For example, social forestry as defined in the PFES policy is a process where all actor groups contribute their financial resources to protect forests and reduce the burden on the state budget (Pham et a. 2013). The Vietnam Forestry Development Strategy (2006-2020) emphasizes an economic and entrepreneurial approach in all forestry actions with local communities and households to advance state development objectives. However, forest "ownership" still only refers to limited forest use rights (except in plantation forests where households have tenure) and little autonomy in decision-making related to their forests (Dang, Turnhout and Arts 2012). An international donor agreed (2019): "Yes, you might see community forestry and social forestry everywhere in policy documents and in the speeches of policy makers, the fact that the state and its state forest enterprises, protected areas and national park and state management boards own and manage more than 85% of forests, mostly the best quality forest in Vietnam while communities and households receive the most degraded forests pose significant challenges for non-state actors to engage in forestry protection and development". By the end of 2018, less than 2% of total forest had been officially allocated to communities (Vietforest, 2018). While there are claims and forms of resistance at different levels, the participation of local people and CSOs in formal SF planning and implementation processes remains limited, despite the participatory rhetoric in policies (from interviews with CSOs in 2019) thus highlighting the state's dominance in the Vietnam assemblage, which effectively dictates territorialization processes to maximize its benefits from Vietnam's forests.

# 4.3. Case study 3: Finding space for Social Forestry within economic development and tenure conflicts in Sabah

The Sabah Forestry Department (SFD), with support from the German Agency for Technical Cooperation, developed and implemented a pilot case for sustainable forest management (SFM) in Deramakot Forest Reserve in 1989 as part of a Forest Stewardship Council (FSC) certification process to address community development issues (UNDP 2008, italics by authors). This pilot approach was considered as a model for SFM, and the acquired FSC certification in 1997 provided impetus for SFD to apply similar 'community forestry' approaches in other FSC forest concessions. The year also marked a significant shift of SFD's core policy approach from industrial logging towards adoption of a SFM strategy and social forestry was institutionalized within this. Under the SFM License Agreement (SFMLA), if there are communities living in or around the forest concession boundaries, the license holders are required to verify community benefits, provide dispute resolution systems and set aside land for community forestry projects within the forest concessions (Toh and Grace, 2005). While acknowledged as an improvement to previous conflicts between industrial logging and local communities, this requirement effectively renders social forestry as a legal and technical obligation for the success of the SFMLA concession. Toh and Grace (2005) argues that it can also be a mechanism for controlling community land use within the forest reserves - a territorialization process where forest managers assign strict boundaries for agriculture plots where only approved tree and cash crops can be cultivated.

Persistence of colonial legislation for economic growth. Conflicts around forest access and rights stem from two key colonial laws: the Sabah Land Ordinance (SLO) of 1930, which recognizes Native Customary Land Rights and includes provisions for the granting of

Native Titles for indigenous individuals and households and Communal Titles for shared access, and the Ladang Act of 1913 which was meant to control shifting cultivation. Both pieces of legislation signaled the North Borneo Chartered Company's (NBCC) intention to settle Native land rights and identify "idle" lands suitable for plantation agriculture (Majid Cooke 2012, Bernard and Bissonnette 2011, Doolittle 2007). The issues of Native rights proved much more difficult for the NBCC however, and this tangled legacy of unresolved rights has led to anxieties, contestation and multiple overlapping claims to forested lands today. Applications for individual Native Titles often take many years to be resolved while logging and plantation licenses are processed quickly<sup>6</sup>. During the peak of palm oil plantation expansion in 2009, there was a reported backlog of 285,000 Native Title applications (Majid Cooke and Toh 2012). The complexities passed on from colonial rule have paved the way for territorialization processes of zoning and the demarcation of land and forests for the long-term state objectives of land commodification and centralized control of natural resources (Lunkapis 2013, Majid Cooke 2013, Doolittle 2007).

In 2010, the Sabah government stopped issuing Native Titles to prioritize communal titles (SUHAKAM 2013). In total, 96 communal titles totaling 61,6201 ha were issued to 13,789 recipients from 271 villages in 15 districts in Sabah (Borneo Post, 2018). The shift to communal titling was built on familiar discourses of low productivity, idle land and poor decisions of natives (Majid Cooke 2012) to push forward the State's strategy for rural development and poverty alleviation (SUHAKAM 2013). An amendment of the SLO in 2009 to include the phrase 'any state land planned by the government for the natives of Sabah' made it "possible for the creation of communal titles in cases where the government plans to develop the land, albeit for the benefit of natives" (Lunkapis 2013, p. 202). The communal title, issued through the Fast Track Land Alienation Program, requires the participants to sign an agreement to transfer their rights to state agents for creation of joint ventures for oil palm plantations on the Native lands (Lunkapis 2013). The communal title came with 'special conditions', including expectations that the land has to be cultivated with commercial plants or trees approved by the Director of Agriculture for the duration of tenure (SUHAKAM 2013, Majid Cooke and Toh 2012). Large tracts of forest land degazetted for Native communal titles were effectively converted to oil palm plantation and the process was deemed to have "violated the rights of natives" (SUHAKAM 2013, p. 93)<sup>7</sup>. The rapid territorialization and de-territorilization of native communal titles legitimized by policy were intended to benefit the state (and convergence of state leaders' and private sector actors' interests) at the expense of the indigenous population.

Potential for a more equitable form of social forestry and native land claims? Within this dynamic context of Native land as the frontier for development objectives, SF in Sabah occupies a fraught space. The active engagement of civil society organizations and regional ASEAN Working Group on Social Forestry (AWG SF) over the past decade has led to the establishment of SASOF with the SFD in 2016. Serving as a multi-stakeholder platform to discuss and align SF activities and to bring new perspectives and ideas to definitions and understandings of SF, SASOF was instrumental in the development of Sabah and National Social Forestry Roadmap (SFD Annual Report 2016). The Roadmap is undergoing consultation and revisions, and in the latest version, SF is defined as "the involvement of indigenous peoples and local

communities on their own initiatives or in collaboration with relevant stakeholders in forest management from the aspects of social, economic, culture and environment for sustainable livelihoods" (MNRE, unpublished draft). At the same time however, social forestry and issues of communal tenure or rights remain conspicuously absent from the Sabah Forest Policy 2018 (SFD 2018) and policy actions only emphasize increased consultations and dialogue with local and Native peoples in forest management, preferential contracting and employment in forest industries, and development of native entrepreneurship (Ibid, p.31). With funding from UNDP, the proposals of CSOs, also SASOF members, to transfer global principles of ICCA (territories and areas conserved by indigenous peoples and local communities) to Sabah provide fresh impetus to integrate Native knowledge and governance practices within SF policies, and is considered by local communities as another mechanism to advance their claims for Native rights (Majid Cooke and Vaz, 2011).

#### 5. Discussion

From our interrogation of the three case studies across Southeast Asia, the framing of problems and proposed solutions in social forestry closely align to what Li (2007) identified as assemblage practices. In particular, we find elements of:

- (1) forging alignments between state, international donors, CSOs and communities on social forestry as a mechanism to govern and control forests, to enable participation and improve local capacity, and as a promise to legal enabling rights. This alignment, though fragile and dynamic, has upheld the social forestry endeavour in its various incarnations for the last four decades;
- (2) the rendering technical of diverse rights and forest governance practices into formal, standardized and enforceable SF programs through bureaucratic and territorialization processes of permitting, licensing and the redrawing of boundaries that potentially excludes more than they include;
- (3) an anti-politics approach to contain the assemblage by integrating non-state actors (and their criticisms and policy concerns) into agents for piloting, implementing and actively supporting formalized social forestry ideals and programs;
- (4) the reassembling of social forestry to promote entrepreneurship and the grafting of win-win-win narratives using market ideals of land reforms for growth (TORA), climate mitigation and ecosystem service payments (REDD+/PES), and value chains (SFM).

These practices are shaped and shaping the interplay of the different public and private actors, their interests and their interactions of contestations and consent in the case study regions – and translate into diverse notions of (in)equity and specific pathways to hold state and private sector accountable to the promises made in the name of social forestry, as we discuss in the following sections.

# 5.1. Actors, interests and the many translations of the notion of equity

In our review of state policies and the stated visions and missions of civil society organizations in relation to social forestry (see Table 2), it is clear that states explicitly maintain authority over forests in the design of formal social forestry mechanisms and its territorialization processes. The three governments in our case studies have had varying roles and prominence in the assemblages over time, with other actors' interests emerging to have influence. Vietnamese state actors evolved from complete authoritarian interests to allow (limited) recognition of traditional and customary rights; Sabah state actors were complicit to external capital interests that benefit the powerful but a short-lived reformist government has state actors partially aligning interests with CSO actors; while the Indonesia government appears fairly consistent in

<sup>&</sup>lt;sup>6</sup> See "Native land issues dominate second day of hearing", Borneo Post, June 6, 2012. The Human Rights Commission of Malaysia (SUHAKAM) conducted a National Inquiry into the Land Rights of Indigenous Peoples from December 2010 to June 2012. The Government of Malaysia has yet to implement any of the recommendations from the report, published in 2013.

<sup>&</sup>lt;sup>7</sup> The communal grants system was recently canceled and replaced with individual Native Titles, an action that was part of the new State Government's election pledges in 2018 (Borneo Post, 2018).

its problematizations and territorialization processes despite various reforms and seemingly high levels of engagement with external actors. However, a not unexpected picture of consensus emerges in the states' problematizations of deficiencies and promises for improvement, highlighting their hard work and graft in forging and maintaining the many elements and social-material connections of the assemblages (Li, 2007). Social forestry is held up as a promise for improving the welfare of local communities and to counter contemporary societal problems such as land scarcity, rural poverty and under-development, but also to advance the States' various interests and claim over forests for economic production and growth.

While equity is explicit in policies such as the 'Equitable Economy (Ekonomi Pemerataan)' in Indonesia, social forestry and TORA programs under this umbrella are de facto mechanisms to formalize forest access and land ownership, or as Scott (2009) argues, to project state power and bring into control "ungoverned regions". In addition, by providing divergent sets of rights and access, the two programs are in actuality likely to increase the potential for conflict: local communities in SF programs have realized that gaining access to state forests also means acknowledging the state's legitimate authority over forests (Fisher et al. 2018, Myers et al. 2017). Local communities in our research sites have expressed preference for TORA titles which promises access to business opportunities and financing, instead of the management burdens that come with social forestry schemes (field interviews, Kapuas Hulu, 2019 and 2020). Territorialization processes associated with formal bureaucracies of SF mechanisms have reduced the initial notion of equity and social-environmental justice related to the devolution of rights and empowerment, to simplistic notions of fair access to markets with restrictive rights and access, and emphasis on economic entrepreneurship and production.

The evolution of social forestry in Vietnam closely mirrors the country's own transformation from a closed socialist state to an open socialist market economy, from a form of socialist forestry where local people provided labor to state forest enterprises to a market-oriented social forestry that channels development aid and financing from PES and REDD+ schemes to local people to manage and rehabilitate state forests. The Government of Vietnam takes pride in its progressive policies, for example being the first country in Southeast Asia to set up a national PES policy, and setting a new precedent official inclusion of ethnic minorities in forest land 'ownership', with the caveats of the limitations in what the term ownership implies. Social forestry continues to be largely framed within principles of economic empowerment as a solution to what are identified as the problems of underdevelopment and rural poverty. However political pressures to increase forestry productivity, mobilize domestic and international investments, and collaborate with enterprises in production to meet centrally planned development targets are the more important State objectives. Equity, thus, is related to the enabling of economic success, limited to 'success' in the current global - and now national - economic framework.

Sabah is still finding its way with social forestry after decades of intensive logging and state-sanctioned 'land grabbing' for oil palm plantation expansion. SFM and the focus on forest certification appears to provide a middle ground to achieving the multiple economic, social and biodiversity goals stated in the latest Sabah Forest Policy 2018. A new state government of the opposition coalition elected in 2018 promised fresh reforms and a progressive outlook, but political machinations in 2020 have led to a return of the former political power and with it, anxiety of potential re-territorialization processes for continued forest exploitation. While economic and biodiversity conservation goals are emphasized through distinct and elaborated actions and plans in the Policy, there is not a single mention of social or community forestry and rights in the policy document. Hope remains in the Sabah Social Forestry Roadmap that has been undergoing deliberations since 2017, but with little expectation that there will be a reverse in the processes of territorialization that has only benefited the powerful.

In our review of civil society organizations (CSOs) strategy

documents, the key civil society actors working in the social forestry space tend to come from the grounded perspectives of environmental justice, forest rights and equity (see Table 2). All the CSOs articulated the detrimental impacts of specific forms of forest conversion and economic development on local people - their exclusion, displacement, increased vulnerability and conflicts - and the environment, and posit the potential role of community or social forestry to address these issues. Notions of empowerment, justice, governance and maintenance of culture and traditional knowledge are evoked, alongside interests and actions to developing forest product enterprises, creating market linkages and enabling private sector partnerships. These latter interests converge with State objectives of social forestry for economic development. CSO actors who have long engaged with policymaking at different government levels, also aimed to promote transformational market reforms that ensure safeguards and secure rights and benefits for local and indigenous communities throughout Southeast Asia (NTFP-EP, RECOFTC). Is this convergence a strategy which CSO actors choose in order to advance their call for reforms or are market-based solutions the only pathway for empowerment in social forestry? Is the notion of equity merely relegated to fair and equal access to markets?

The relationship between the State and CSOs is more dynamic. In our case studies, policymakers in both Indonesia and Sabah have found ways to work with CSOs to advance their social forestry objectives while Vietnam has largely chosen not to engage with non-State actors, except for select donor institutions contributing finance to the forest sector. CSOs in Vietnam are also more constrained in their activities, often scrutinized by the government. The Indonesia Working Group on Social Forestry started as an independent group of practitioners, researchers and activists with diverse interests in varying issues related to social forestry such as community development, indigenous rights and forest conservation, and in its institutionalization as POKJA PPS, its members are now deployed as experts to facilitate the establishment of formal social forestry schemes to help meet the Government target. SASOF on the other hand is a collaboration between the state (SFD), civil society organizations and academia, first initiated after an ASEAN social forestry network conference held in Sabah in 2014, and formalized in 2016. Their agreed mandate was to develop and promote social/community forestry; customary and local stewardship, governance and management of forests and other natural resources; and sustainable community enterprises and livelihoods. The CSO members of SASOF are advocating for ICCA principles of recognition of native customary knowledge and rights, while conscious of the limits within Sabah's policies. In both case studies, the CSOs are embedded within social forestry policy processes, but it is not yet clear whether their voiced concerns of social justice are fully recognized and distinct within the discourses of policy making or whether they are co-opted into a political process to enhance State legitimacy by enabling/facilitating the implementation of pre-determined social forestry models that have generally not benefitted local forest communities in an equitable manner (Bong et al. 2019, Moeliono et al. 2017, Sahide et al. 2020). State actors engaged in these networks have interests of varying convergences with CSO and international actors while maintaining formal allegiance to state goals, and these interactions are perhaps the glue to holding together the assemblages. Whether the belief that equity can be found within the formal structures of Indonesia's SF schemes or the call for recognition equity is heard in Sabah, CSOs remain the only actors likely to advocate for such claims.

# 5.2. Public interests, private invisibilities and accountability

In our examination of the social forestry assemblage in the case studies, one group is conspicuously silent: market linkages, tourism, NTFP production and community-based enterprises are mentioned in all institutional and policy objectives, and the private sector actor is invoked everywhere, yet there is little visible engagement of the private sector actors in social forestry policy processes. CSOs provide some

definition of their private sector of interest: NTFP-EP identifies "ethical domestic, regional and international markets for NTFPs and creative industries" as a target for the community produced NTFP products (NTFP-EP 2016, p. 17) and RECOFTC identifies "both large enterprises and SMEs [small medium enterprises] and responsible investors" as business and investment partners for the community based enterprises (RECOFTC, 2018, p. 32). However, the private sector actor or market linkage is somewhat hidden in policies that focus only on creating the 'enabling' conditions: permits for the Indonesian formal scheme of Village Forests (Hak Pengelolaan Hutan Desa) can only be requested by a village cooperative or enterprise (Fisher et al. 2018, Moeliono et al. 2015), signaling that village forests are to be managed to generate revenues, while Vietnamese policies emphasize increasing productivity, creating flexible models of collaboration with enterprises and promoting specialized forest production zones. The belief that markets and entrepreneurship is a solution to the problematizations of conflict, poverty and increasing vulnerability of local forest communities and forest degradation is a prevalent discourse - including throughout ASEAN level policy agreements – and this possibly assigns power to the distal private sector actors while responsibilities of this actor group remain undefined in social forestry policy processes. Our case study of Sabah highlights how the fast-track process of communal titles were designed to make land available to private or public-private development interests, and the return of old political power may yet re-enable these convergences. Other studies have shown how states do clearly represent private sector interests in climate and forest policy decisions (Brockhaus et al. 2014, Ross 2001), and this is a critical area for further study, and theoretical and methodological advances. The evidence on outcomes from increasing market access and entrepreneurship is still sparse and Governments often do not have systematic monitoring in place for assessing multi-dimensional outcomes of SF despite their rhetoric. This raises interesting questions on who would be held accountable if the impacts are less than desirable or even negative: the local communities for lacking necessary skills to properly engage with markets, the State and their 'enabling' policies, the CSOs that promote the partnerships, or the private sector actors who are absent and hence do not enable scrutiny for society in social forestry schemes to hold them accountable.

#### 6. Conclusion

The endeavor of social forestry has forged alignment across state, civil society and international donor actors, and with local communities. Social forestry was initially framed around objectives of equity, empowerment and devolution of rights to local communities to continue their stewardship and governance practices over the forests, while providing a basis for their livelihoods and development. Social forestry was later re-assembled and entangled with the initiatives of agrarian reform in Indonesia, SFM in Sabah and PES and REDD+ narratives in Vietnam, with the underlying narrative that entrepreneurship and fair access to markets is the equitable solution towards economic empowerment and prosperity. Meanwhile, the private sector as a desired new actor in the reassembling of SF is absent from the public discourses but may exercise most power in profiting from economic exploitation of forests and land. While there are voices calling for social justice within the assemblage, these tend to be muted in the anti-politics approach of being assimilated into the practice or implementation of policy as in Indonesia and Sabah, or in their exclusion from the policy process altogether as in Vietnam.

Our work contributes to the as yet limited scholarship deploying assemblage approaches in forestry. The case study findings suggest that a multi-dimensional notion of equity should involve recognition of the diverse governance practices and institutions of local forest communities, instead of the imposition of formal and structured social forestry models from the top-down and discourses inherited from colonial times. Recognition of this diversity would mean that social forestry models has to move current territorialization processes of exclusion to consider

property arrangements that may involve different constellations and notions of property and rights and accommodate meanings of forest and landuse beyond economic production values (Bong et al. 2019, Sikor and Müller, 2009). While the principles of ICCA being advocated in Sabah could be a start, social forestry in Southeast Asia is far away from meeting this notion of recognition equity. Most SF schemes, rendered technical through formal bureaucracy and territorialization processes that lead to exclusions and stricter rules – and enforcement – on forest use, may in fact reinforce inequities as the poorest are often most affected in these forms of decentralization (Lund et al. 2018).

We suggest that future research focus on impact assessments of social forestry in Southeast Asia, beyond simplistic metrics of hectares and incomes but by asking (and measuring) who benefits from social forestry shemes' implementation. Crucial will be an understanding of how SF policies and practices of territorializations have affected inclusions and exclusions over rights, participation and access over time, have influenced notions of local equity, changed or reinforced contestations and differentiation in forest access and use; and of considerations of power relations in governance practices and market linkages in dynamic forest landscapes. Application of the assemblage approach has allowed for a more nuanced understanding of the heterogenous, contesting and intersecting interests within the social forestry, and this is particularly critical in Southeast Asian rural forest community-state contexts where pervasive State power, do not always allow for conflicting and diversity of views.

#### CRediT authorship contribution statement

Grace Y. Wong: Conceptualization, Methodology, Writing - original draft, Writing - review & editing. Moira Moeliono: Conceptualization, Writing - review & editing. Indah W. Bong: Writing - review & editing. Pham Thu Thuy: . Muhammad A.K. Sahide: . Daisuke Naito: . Maria Brockhaus: Conceptualization, Methodology, Writing - review & editing.

#### References

Agrawal, A., Angelsen, A., 2009. Using community forest management to achieve REDD + goals. In: Angelsen, A. (Ed.), Realising REDD+: National Strategy and Policy Options. CIFOR, Bogor Indonesia.

Anderson, B., McFarlane, C., 2011. Assemblage and geography. Area 43 (2), 124–127.
Astuti, R., McGregor, A., 2017. Indigenous land claims or green grabs? Inclusions and exclusions within forest carbon politics in Indonesia. J. Peasant Stud. 44 (2), 445–466

Bernard, S., Bissonnette, J.F., 2011. Oil palm plantations in Sabah: agricultural expansion for whom. In: Bissonnette, J.F., Bernard, S., DeKoninck, R., (Eds). Borneo transformed: Agricultural expansion on the Southeast Asian frontier, pp.126-151. Singapore, National University of Singapore Press.

Bong, I.W., Moeliono, M., Wong, G.Y., Brockhaus, M., 2019. What is success? Gaps and trade-offs in assessing the performance of traditional social forestry systems in Indonesia. Forest Soc. 3, 1–21.

Borneo Post (2012) Native land issues dominate second day of hearing, published June 6, https://www.theborneopost.com/2012/06/06/native-land-issues-dominate-second-day-of-hearing/ accessed 30/9/2019.

Borneo Post, 2018. Sabah abolishes communal grants, replaces it with individual land titles, published August 9, https://www.theborneopost.com/2018/08/09/sabah-abolishes-communal-grants-replaces-it-with-individual-land-titles/, accessed 30/9/2019

Brockhaus, M., Di Gregorio, M., Mardiah, S., 2014. Governing the design of national REDD+: an analysis of the power of agency. Forest Pol. Econ. 49, 23–33.

Cleary, M., 2005. Managing the forest in colonial Indochina c. 1900-1940. Modern Asian Studies, 39(2), 257–283.

Clement, F., Amezaga, J.M., 2008. Linking reforestation policies with land use change in northern Vietnam: why local factors matter. Geoforum 39, 265–277.

Cole, R., Brockhaus, M., Wong, G.Y., Moeliono, M., Kallio, M., 2019. Agency in development, market and forest conservation interventions in Lao PDR's northern uplands. Southeast Asian Studies 8 (2), 173–202.

Cole, R., Wong, G., Bong, I.W., 2017. Implications of the ASEAN Economic Community (AEC) for trans-boundary agricultural commodities, forests and smallholder farmers. CIFOR Infobrief no. 178. CIFOR, Indonesia.

Cons, J., Eilenberg, M., 2019. Introduction: On the new politics of margins in Asia: mapping frontier assemblages. In Cons and Eilienberd (eds). In: Frontier assemblages: the emergent politics of resource frontiers in Asia. Wiley, Oxford, pp. 1–18.

- Cummins, A., Yamaji, E., 2019. To see invisible rights: Quantifying Araman informal tenure and its immediate relationship with Social Forestry in Central Java, Indonesia. For. Soc. 3 (2), 193–201.
- Dang, T.K.P., Turnhout, E., Arts, B., 2012. Changing forestry discourses in Vietnam in the past 20 years. Forest Pol. Econ. 25, 31–41.
- De Jong, W., 2012. Discourses of community forestry. In: Arts, B., van Bommel, S., Ros-Tonen, M., Verschoor, G., ed. Forest-people interfaces: from local creativity to global concern, pp. 107-120. Netherland: Wageningen Academic Publishers.
- De Royer, S., Van Noordwijk, M., Roshetko, J.M., 2018. Does community-based forest management in Indonesia devolve social justice or social costs? Int. For. Rev. 20 (2), 167–180.
- Doolittle, A.A., 2001. From village land to "native reserve": changes in property rights in Sabah, Malaysia, 1950–1996. Human Ecol. 29 (1), 69–98.
- Doolittle, A.A., 2007. Native land tenure, conservation, and development in a pseudo-democracy: Sabah, Malaysia. J. Peasant Stud. 34 (3–4), 474–497.
- Erbaugh, J.T., 2019. Responsibilization and social forestry in Indonesia. For. Pol. Econ. 109
- FKKM (Forum Komunikasi Kehutanan Masyarakat) (2010) Warta Edisi Augustus 2010, https://drive.google.com/file/d/0B48XF909w0LwcXBmb2RlUU5ZNnc/view, accessed 30/9/2019
- FKKM, 2017. Warta Edisi November 2017, https://drive.google.com/file/d/ 1YGEo78DHIDfhXzNGkAmtNMDBi5E6uCWK/view, accessed 30/9/2019.
- Fisher, M.R., Moeliono, M., Mulyana, A., Yuliani, E.L., Adriadi, A., Judda, J., Sahide, M. A.K., 2018. Assessing the new social forestry project in Indonesia: recognition, livelihood and conservation? Int. Forest. Rev. 20 (3), 346–361.
- Gilmour, D., 2016. Forty years of community-based forestry: a review of its extent and effectiveness. Food and Agriculture Organizations of the United Nations, Rome, Italy
- Hajjar, R., Badini, O.S., Kozak, R.A., 2016. Promoting small and medium forest enterprises in national REDD + strategies: a global analysis of enabling environments. Clim. Pol. 731–763.
- Kelley, L.C., Peluso, N.L., Carlson, K.M., Afiff, S., 2020. Circular labor migration and land-livelihood dynamics in Southeast Asia's concession landscapes. J. Rural Stud. 73, 21–33
- KSP (Kantor Staf Presiden), 2016. Strategi Pencapaian Target Program Prioritas Reforma Agraria https://hanibalhamidi.files.wordpress.com/2016/09/naskah-stranaspelaksanaan-reforma-agraria-ksp.pdf, accessed 2/10/2019.
- Leach, M., 2002. Plural perspectives and institutional dynamics: challenges for community forestry. In: Oglethorpe J. (ed) Adaptive Management: From Theory to Practice, pp. 67-82. SUI Technical Series 3 IUCN, Gland Switzerland.
- Li, T.M., 2007. Practices of assemblage and community forest management. Econ. Soc. 36 (2), 263–293.
- Lund, J.F., Rutt, R.L., Ribot, J., 2018. Trends in research on forestry decentralization policies. Curr. Opin. Environ. Sustain. 32, 17–22.
- Lunkapis, G.J., 2013. Confusion over land rights and development opportunities through communal titles in Sabah, Malaysia. Asia Pacific Viewpoint 54 (2), 198–205.
- Macqueen, D., 2013. Enabling conditions for successful community forest enterprises. Small-scale Forestry 12 (1), 145–163.
- Maharani, C.D., Moeliono, M., Wong, G.Y., Brockhaus, M., Carmenta, R., Kallio, M., 2019. Development and equity: a gendered inquiry in a swidden landscape. Forest Pol. Fcon. 101, 120–128.
- Majid Cooke, F., 2012. In the name of poverty alleviation: Experiments with oil palm smallholders and customary land in Sabah, Malaysia. Asia Pacific Viewpoint 53 (3), 240–253.
- Majid Cooke, F., 2013. Constructing rights: Indigenous peoples at the public hearings of the national inquiry into customary rights to land in Sabah, Malaysia. SOJOURN: Journal of Social Issues in Southeast Asia 28 (3), 512–537.
- Majid Cooke, F., Toh, S.M., 2012. Indigenous Peoples and Access to Customary Lands: A Question of Rights in Contemporary Sabah. Background paper for the SUHAKAM National Inquiry on Land Rights for Indigenous Peoples.
- Majid Cooke, F., Vaz, J., 2011. A Review of Indigenous and Community-Conserved Areas in Sabah. Global Diversity Foundation for the Sabah Biodiversity Centre, Bornean Biodiversity and Ecosystem Conservation Project II, Kota Kinabalu.
- MARD (Ministry of Agriculture and Rural Development Vietnam), 2017. The new Forestry Law will impact on 16,2 million hectare of forest, https://www.mard.gov. vn/Pages/bo-nnptnt-luat-lam-nghiep-se-tac-dong-den-16-2-trieu-ha-dat-rung.aspx, accessed 12/12/2019.
- McElwee, P., 2011. Who should manage the land? Common property and community responses in Vietnam's shifting uplands. In: Sikor, T. (Ed.), Upland Transformations in Vietnam. National University of Singapore Press, Singapore, pp. 75–91.
- Moeliono, M., Pham, T.T., Bong, I.W., Wong, G.Y., Brockhaus, M., 2017. Social Forestry why and for whom? A comparison of policies in Vietnam and Indonesia. Forests Soc. 2 (1), 1–20.
- Moeliono, M., Pham, T.T., Le, N.D., Brockhaus, M., Wong, G., Kallio, M., Nguyen, D.T., 2016. Local governance, social networks and REDD+: lessons from swidden communities in Vietnam. Human Ecol. 44, 435–448.
- Moeliono, M., Mulyana, A., Adnan, H., Manalu, P., Yuliani, L., Balang, 2015. Village forests (hutan desa): empowerment, business or burden? Agroforestry and Forestry in Sulawesi no. 51. ICRAF- Southeast Asia Regional Office, Bogor, Indonesia.
- MOEF (Ministry of Environment and Forestry), 2018. State of Indonesia's Forest 2018. Jakarta, MOEF.
- Myers, R., Intarini, D., Sirait, M.T., Maryudi, A., 2017. Claiming the forest: Inclusions and exclusions under Indonesia's 'new' forest policies on customary forests. Land Use Pol. 66, 205–213.
- Nel, A., 2017. Contested carbon: carbon forestry as a speculatively virtual, falteringly material and disputed territorial assemblage. Geoforum 81, 144–152.

Ngo, T.D., Mahdi, 2017. Targeting deforestation through local forest governance in Indonesia and Vietnam. In: Redefining Diversity & Dynamics of Natural Resources Management in Asia, pp. 273–288. Elsevier.

- NTFP-EP, 2016. Linking people and forests. Strategic Direction 2016-2019. Manila, NTFP-EP, https://ntfp.org/wp-content/uploads/2017/07/Strat-Paper-2016-0503177LR.pdf, accessed 3/7/2019.
- Peluso, N.L., 1996. Fruit trees and family trees in an anthropogenic forest: Ethics of access, property zones, and environmental change in Indonesia. Comparative Stud. Soc. Hist. 38 (3), 510–548.
- Peluso, N.L., 1992. Rich Forests, Poor People: Resource Control and Resistance in Java. University of California Press, Berkeley.
- Peluso, N.L., Purwanto, A.B., 2018. The remittance forest: turning mobile labor into agrarian capital. Singap. J. Trop. Geogr. 39 (1), 6–36.
- Peluso, N.L., Vandergeest, P., 2001. Genealogies of the political forest and customary rights in Indonesia, Malaysia, and Thailand. J. Asian Stud. 60 (3), 761–812.
- Peluso, N.L., Vandergeest, P., Potter, L., 1995. Social aspects of forestry in Southeast Asia: a review of postwar trends in the scholarly literature. J. Southeast Asian Stud. 26 (1), 196–218.
- Pham, T.T., Di Gregorio, M., Carmenta, R., Brockhaus, M., Le, D.N., 2014. The REDD+ policy arena in Vietnam: participation of policy actors. Ecol. Soc. 19 (2), 22.
- Pham, T.T., Moeliono, M., Nguyen, T.H., Nguyen, H.T., Vu, T.H., 2012. The context of REDD+ in Vietnam: Drivers, agents and institutions. CIFOR Occasional Paper 75. Bogor, Indonesia: Center for International Forestry Research (CIFOR).
- Pham, T.T., Bennet, K., Vu, T.P., Brunner, J., Le, N.D., Nguyen, D.T., 2013. Payments for forest environmental services in Vietnam: From policy to practice. Occasional Paper 93. Bogor, Indonesia: CIFOR.
- Pham, T.T., Hoang, M.H., Campell, B., 2008. Pro-poor payments for environmental services: challenges for the government and administrative agencies in Vietnam. Public Admin. Dev. 28 (5), 262–272.
- RECOFTC, 2017. Social forestry and climate change in the ASEAN region: Situational analysis 2016. RECOFTC The Center for People and Forests, Bangkok, Thailand.
- RECOFTC, 2014. Current status of social forestry in climate change mitigation and adaptation in the ASEAN region: Situational analysis 2013. RECOTC, Bangkok, Thailand.
- RECOFTC, 2018. Resilient Landscapes: Empowered Communities, Strengthened Institutions and Shared Prosperity (Strategic Plan 2018-2023). RECOFTC. The Center for People and Forests, Bangkok, Thailand.
- Resosudarmo, I.A.P., Tacconi, L., Sloan, S., Hamdani, F.A.U., Alviya, I., Muttaqin, M.Z., 2019. Indonesia's land reform: Implications for local livelihoods and climate change. For. Pol. Econ. 108.
- Rigg, J., 2003. Evolving rural-urban relations and livelihoods. In Chia L.S. (ed.) Southeast Asia Transformed: A Geography of Change, pp. 231–256. ISEAS–Yusof Ishak Institute Singapore.
- Ross, M.L., 2001. Timber booms and institutional breakdown in Southeast Asia.

  Cambridge University Press.
- SFD (Sabah Forest Department), 2016. Sabah Forestry Annual Report 2016. Sandakan, Sabah, http://www.forest.sabah.gov.my/docs/ar/ar2016.pdf, accessed 30/9/2019.
- SFD (Sabah Forest Department), 2018. Sabah Forest Policy 2018. Sandakan, Sabah.
- Sahide, M.A.K., Fisher, M.R., Erbaugh, J.T., Intarini, D., Dharmiasih, W., Makmur, M., Maryudi, A., 2020. The boom of social forestry policy and the bust of social forests in Indonesia: developing and applying an access-exclusion framework to assess policy outcomes. For. Pol. Econ. 120, 102290.
- Sahide, M.A.K., Giessen, L., 2015. The fragmented land use administration in Indonesia analysing bureaucratic responsibilities influencing tropical rainforest transformation systems. Land Use Pol. 43, 96–110.
- Seymour, F., Rutherford, D., 1993. Contractual agreements in the Java Social Forestry program, in: Fox, J. (Ed), Legal Frameworks for Forest Management in Asia: Case studies of Community/State Relations. Environment Program Occasional Paper no. 16. East West Center, Honolulu, USA.
- Sikor, T., Müller, D., 2009. The limits of state-led land reform: an introduction. World Dev. 37 (8), 1307-1316.
- Sikor, T., Nguyen, T.N., 2011. Realizing Forest Rights in Vietnam: Addressing Issues in Community Forest Management. RECOFTC, Hanoi.
- Siscawati, M., Banjade, M.R., Liswanti, N., Herawati, T., Mwangi, E., Wulandari, C., Tjoa, M., Silaya, T., 2017. Overview of forest tenure reforms in Indonesia. Working Paper 223. CIFOR, Bogor, Indonesia.
- Skutsch, M., Turnhout, E., 2018. How REDD+ is performing communities. Forests 9 (10), 638
- SUHAKAM, 2013. Report of the National Inquiry into the Land Rights of Indigenous Peoples. Kuala Lumpur, Human Rights Commission of Malaysia, https://nhri.ohchr.org/EN/Themes/BusinessHR/Business%20Womens%20and%20Childrens% 20Rights/SUHAKAM%20BI%20FINAL.CD.pdf, accessed 30/9/2019.
- Thu, T.P., Moeliono, M., Brockhaus, M., Wong, G.Y., Le, N.D., in press. The politics of swidden: A case study from Nghe An and Son La in Vietnam. Land Use Policy, available online 17 March 2018.
- Toh, S.M., Grace, K.T., 2005. Case Study: Sabah Forest Ownership. Report produced for the Food and Agriculture Organization. http://www.fao.org/3/j8167e/j8167e10. pdf, accessed 3/7/2019.

UNDP, 2008. Malaysia Sustainable Community Forest Management in Sabah. Kuala Lumpur, UNDP Malaysia. https://theredddesk.org/sites/default/files/resources/pdf/2012/forest\_mgmt\_final.pdf, accessed 3/7/2019.

Vietforest, 2018. Rationales to enhance community forestry. Available at: http://vietfores.org.vn/tin-tuc/mot-ly-chinh-de-phat-trien-lam-nghiep-cong-dong-o-viet-nam/, accessed 1/12/2019.

Wong, G.Y., Luttrell, C., Loft, L., Yang, A.L., Pham, T.T., Naito, D., Assembe-Mvondo, S., Brockhaus, M., 2019. The narratives in REDD+ benefit sharing: examining evidence within and beyond the forest sector. Clim. Pol. 19 (8), 1038–1051.